Author’s response to reviews

Title: Minimum volume standards in day surgery: a systematic review

Authors:
Michal Stanak (michal.stanak@protonmail.com)
Christoph Strohmaier (christoph.strohmaier@aihta.at)

Version: 1 Date: 01 Jul 2020

Author’s response to reviews:

Dear Editor,

In your last email, I was asked to do 3 things:

1) rename Objective to Background in the abstract section,
2) rename Introduction to Background in the main body of the text, and
3) move the list of abbreviations before the Declarations section.

Points 1 and 3 were actually already corrected in the previous revision and so did not change those two. Point 2 was changed according to your instructions.

Best regards,
Michal Stanak